



Davies, M. A. M., Balai, E., Adams, J., Carter, J-H., Judge, A., Newton, J. L., & Arden, N. K. (2017). The consultation of rugby players in codeveloping a player health study: feasibility and consequences of sports participants as research partners. *Research Involvement and Engagement*, *3*, 8. https://doi.org/10.1186/s40900-017-0055-1

Publisher's PDF, also known as Version of record

License (if available): CC BY

Link to published version (if available): 10.1186/s40900-017-0055-1

Link to publication record in Explore Bristol Research PDF-document

This is the final published version of the article (version of record). It first appeared online via BMC at https://researchinvolvement.biomedcentral.com/articles/10.1186/s40900-017-0055-1. Please refer to any applicable terms of use of the publisher.

University of Bristol - Explore Bristol Research General rights

This document is made available in accordance with publisher policies. Please cite only the published version using the reference above. Full terms of use are available: http://www.bristol.ac.uk/pure/about/ebr-terms

Open Access



The consultation of rugby players in codeveloping a player health study: feasibility and consequences of sports participants as research partners

Madeleine A.M. Davies^{1,2}, Edward Balai³, Jo Adams⁴, John-Henry Carter⁵, Andrew Judge^{1,6}, Julia L. Newton² and Nigel K. Arden^{1,2,6*}

* Correspondence: nigel.arden@ndorms.ox.ac.uk ¹Oxford NIHR Musculoskeletal Biomedical Research Unit, University of Oxford, Oxford, UK ²Arthritis Research UK Centre for Sport, Exercise and Osteoarthritis, University of Oxford, Oxford, UK Full list of author information is available at the end of the article

Plain English summary

Many funding bodies within the United Kingdom and globally have encouraged public involvement in research. The Department of Health has also called public involvement a sign of good research. Despite the wide acceptance of public involvement improving many aspects of research, from its design to its communication, involvement has varied levels of implementation across different fields of research. Sports people have rarely been involved in research, partly as this research tends not to be funded by mainstream funding bodies. This may lead to a lower research quality, not founded in player ('service user') experiences.

When creating a study of former rugby player health, we were very keen to involve rugby players, understand their thoughts on player health, and their playing experiences. This article explains how rugby players were involved in several ways, but mainly in group discussions during the design stage. These groups helped to inform our study's aims and questionnaire, ensure the questionnaire would capture player experiences and answer questions relevant to players, that they would like to understand after their participation in rugby.

We found that these groups were easy to arrange, and that in only one session with each group, we were given many ideas of how to improve the questionnaire and study. We believe that other studies in sports should involve sports people, and that this is a useful activity that will change data collection forms and processes, improving the research, helping researchers, and making studies more suitable for players who take part in them.

Abstract

Background Patient and public involvement ('involvement') in the UK has increased in accordance with funding requirements, patient-centered health policy initiatives and reporting of the positive impact of involvement for those involved, research and researchers. However, involvement has not been implemented equally across all disease areas and populations. The aim of this process was to involve rugby players across the research cycle of a player health study, ensure the study is player-centred, and that players had approved and informed the design of the study and its questionnaire from their playing experiences.

Methods Two group discussions were undertaken with current students who were (Continued on next page)



© The Author(s). 2017 **Open Access** This article is distributed under the terms of the Creative Commons Attribution 4.0 International License (http://creativecommons.org/licenses/by/4.0/), which permits unrestricted use, distribution, and reproduction in any medium, provided you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons license, and indicate if changes were made. The Creative Commons Public Domain Dedication waiver (http://creativecommons.org/ publicdomain/zero/1.0/) applies to the data made available in this article, unless otherwise stated.

(Continued from previous page)

playing rugby at a Collegiate University. All male and female University rugby players and two College rugby teams were approached to become involved. Sessions were chaired by a player-lead using a topic guide and were audio-recorded and transcribed. Player suggestions were extracted by the player-lead and discussed within the study team for inclusion in the player health study and its guestionnaire. Results Players readily engaged with the sessions and made many contributions to the development of the study and the questionnaire. Players discussed whether certain topics were being collected satisfactorily, and whether the guestionnaire would encompass their playing experiences or that of other players. Players suggested where answers might be less reliable, and ways in which this could be improved. Players recommended additions to the guestionnaire, and guestioned researchers on the choice of language, motivation for question inclusion and if measures were standardised or novel. Alterations were made to the questionnaire based on suggestions, where these were agreed by the study team. **Conclusions** Involving a group of players in the design of a player health study and guestionnaire was not an arduous process and was rewarding for researchers. The

questionnaire was not an arduous process and was rewarding for researchers. The process resulted in numerous alterations to the questionnaire and its functionality, which may improve response rate, the experience of players participating in the player health study, and their ability to report relevant information aligned with their previous experience. Player involvement in research was feasible to implement and improved not only the questionnaire, but also researcher confidence in the project and player experiences being accurately captured and leading a reliable data collection processes in a population with the potential for cultural bias to affect the ascertainment of health, pain and injury.

Keywords: Player, Involvement, User, Design, Public, Health, Consultation, Rugby

Background

Public involvement ('involvement') is defined as "research being carried out 'with' or 'by' members of the public, rather than 'to', 'about' or 'for' them" [1] and is now widely seen as a component of good research practice, increasing its quality, relevance and accountability [2–4]. Involvement has also recently become a prerequisite for NIHR funding [5] and national guidelines to assist with the integration of involvement into health service research have been developed by advisory group INVOLVE [1].

Health policy has been central to the growth of involvement in research across several developed countries, with the National Health Service Research and Development Strategies calling for its integration into the healthcare system and healthcare research [6]. Involvement is encouraged by the Department of Health, who state that 'relevant service users and carers or their representative groups should be involved wherever possible in the design, conduct, analysis and reporting of research' [7]. The term 'Public and Patient Involvement and Engagement' includes members of the public, as citizens with the potential for health service utilisation, and also service users who are current patients or their relatives and carers, who may have a specific investment in improving services and research, and ensuring their views and experiences are included in decision-making within healthcare.

The need to strengthen reporting and measurement of impact within involvement has been previously discussed [4, 8]. Tokenistic involvement, whereby the mandatory inclusion of involvement on funding applications, and as a sign of good practice, may contribute to involvement being considered a 'tick box exercise', has been previously considered [4, 9]. Strengthening the reporting of involvement and encouraging publication has been considered as one method of preventing conscious or subconscious tokenistic involvement within research [4]. The importance of working with a group that will provide the knowledge and perspective that research teams would like from members of the public and may be missing from their team has also been discussed as central to successful involvement [1, 8].

Involvement can be appropriate for all stages of the research process [5] including research bids [10], naming and branding studies [11], developing interventions [12, 13], trials [2] or placebo design [14]. Despite the recommendation of involvement across all stages of research and for all relevant patients or service users, involvement has not been implemented equally across different disease areas and populations with an International bibliometric review of involvement in health research demonstrating that involvement has traditionally being undertaken more so with black and minority ethic and indigenous populations, and most prominently research areas of mental health, indigenous population health and cancer, followed by sexual health, children and parenting, and diet, obesity and diabetes [15]. This clustering amongst populations and illness areas may be as these groups are more regularly being involved in healthcare and research, the chronic nature of these conditions permitting the rapport and the opportunity to undertake involvement activities or that these endeavours are more regularly documented [15] or undertaking progressive and impactful involvement.

The public has been described by INVOLVE as including 'patients, potential patients, carers and people who use health and social care services as well as people from organisations that represent people who use services' [1]. In research that is not focusing on healthcare services or patients, the importance of involving those specific populations and understanding their experiential knowledge may be equally valuable to the research process, those involved, and researchers.

Despite the growth of involvement and its widespread adoption across many different disease areas and research stages [15], there has been limited application of involvement to studies involving sporting populations, either to involve them as citizens, or a specific population with the potential for increased service use following sports participation. With the benefits of involvement for researchers, members of the public and research quality now well-documented [1, 8, 16, 17], there is a need to emphasise the ease of integrating involvement into studies involving sports participants, and to encourage these fields to adopt the good practice of other areas of healthcare science and medicine. This will not only increase research relevance, quality and potentially recruitment, by providing accessible patient-facing documents, but also give voice to those who will ultimately be affected by research outcomes, as service users with a specific identity that may affect treatment decisions and their perception of clinical processes and judgments.

The motivation for involvement in research has previously been described as epistemological, moralistic and consequentialist [18]. Epistemological motivation for involvement is the experiential knowledge that can be bought to the project, its processes and researchers by involving specialists in that environment, such as patients, carers and service-users. Moralistic motivation has been described as the involvement of the public in research for moral, ethical or public right [18]. Research that is publiclyfunded or charity-funded should be involving those who are funding this work in order to improve transparency and that those likely to be ultimately affected by the outcomes of research, should be contributing to this research. The consequentialist or effectiveness argument describes how research can be enriched with the involvement of members of the public and that there can be improvements in its quality, relevance and impact with involvement [19].

Experiential knowledge in rugby union

The experiential knowledge of sports participants is extensive and unique to their specific sport's environment. Sport-specific terminology may represent a unique dialect and shared values of team members have been previously discussed as contributing to a sport-specific subculture or individual society [20]. The experiential knowledge of players about their own environment, sport's ethos and attitude to health and research are unique, and involving players in research is the only way to ensure player experiences are truly encompassed and represented within a research study.

Involvement may be particularly beneficial in areas where groups may be hostile to services or service providers [8]. Due to recent widespread scrutiny of player health in rugby union [21–23], understanding players' attitudes to research and researchers will assist with anticipating player response and mitigating players concerns surrounding the portrayal of a sport, where player values may result in an unwillingness to cast the sport in a negative light [20].

In addition to increasing the quality and relevance of research and meeting demands of public accountability and transparency, the unique sporting subculture of rugby union has been previously acknowledged as possessing a cultural non-acceptance of pain and injury [24, 25]. Involving players as co-researchers, collaborators and consultants throughout various stages of this research project was hoped to increase the accuracy of data collection and self-reported outcomes of pain and disability, within a population historically appreciated to consciously disregard pain and injury [25].

Whilst not previously having been documented as involved in collaborative research, there is precedent for involving the most relevant members of the public with a specific project. It has been recommended to consider the knowledge and perspective that you are looking for from members of the public when working with them with a research project [1]. Involvement has been effectual in scenarios where specific populations have higher levels of health inequality, but also may have contrasting worldviews or cultural values that either contribute to this, or impede progress to reduce inequalities [26, 27].

Can sports people be considered marginalised in healthcare?

Research within sports science exploring physiology, psychology, nutrition or increasing elite performance and health, or research within sports medicine exploring injury, pain and pathology, have thus far been extremely limited in their involvement of members of the public, or sports participants.

Sports participants may not traditionally be viewed as disempowered or marginalised, however could be viewed as such when considering their capacity to make individual and unconstrained choices with regard to their health and healthcare, such as surgery or return to play decisions. Within elite sports environments, numerous factors such as squad availability, league placement, upcoming fixtures and player motivation may influence healthcare decisions. The healthcare decision-making process may encompass a team of support staff with varied interests, including coaches and managers, institutional healthcare providers such as team physicians, medical officers or national institutes for sports. The complexities of return to play decisions in elite sports, such as low-compliance with return-to-play procedures, and whether a full recovery from injury is needed in order to train or compete have been previously considered, with 12% of club personnel within New Zealand rugby clubs believing a full recovery was essential for a return to competition [28–30].

Amateur or recreational players may not experience the same pressures as professional athletes in terms of managing injury in the professional sporting environment, but this specialist environment also often provides the opportunity for novel and comprehensive injury treatment and management. Recreational athletes who are otherwise healthy members of the public requiring generally acute primary care utilisation for a specific injury may not be treated by clinicians with expertise in Sport and Exercise Medicine, who are aware of their potential environmental pressures or sporting demands. In comparison to other service users, recreational athletes may also be not as actively involved in patient-centred care or ensuring their needs are adequately met, possibly as a result of shorter admissions, preventing athletes becoming the traditionally more involved chronic morbidity populations [15]. In addition to limited evidence of recreational athlete involvement in health service provision, there has also been little historical involvement as co-researchers or collaborators within research studies.

This article documents the different forms of involvement of current and former rugby players, as a specific group of service users, across various stages of the research cycle in an English study of former rugby player health and the feasibility and benefits of integrating player involvement into sporting research practices. A primary consideration in the methodology of this player health study was significant player involvement and research being player-centered.

Involvement during the study took place in several ways:

- 1. Collaborative study management roles
- 2. Individual scoping discussions with current and former, recreational and elite players during the design phase and before the dissemination of results
- 3. Consultancy in 'fora' (plural of forums) informing study design, questionnaire content and recruitment methodology
- 4. Collaborators in testing data collection platforms for ease of use and test-retest validity reliability

This article will focus particularly on the fora: the format, suggestions and outcomes of this, and ways in which involvement may be better implemented in sporting populations in the future. The authors are aware of discussion surrounding the classification as members of public as 'patients', 'participants', 'consumers', 'collaborators', 'co-researchers' and 'consultants' depending on the research area and level of involvement. Within this article members of the fora involved in the design stages of the project will be referred to as 'players' as they were approached due to their experiential knowledge as a result of this status.

The aim of the fora were to involve current players in the design and development of an accessible and player-focused questionnaire, with limited bias resulting from the sport's sub-culture and values.

Method

The player health study was a cross-sectional questionnaire study with retrospective data collection, the results of which will be reported elsewhere.

The setting of the fora was a high-ranking Collegiate University. The entirety of the University's male and female rugby teams, and two specific College rugby teams were approached through mailing list administrators to attend one of two sessions being coordinated at the aforementioned Colleges. Details of the sessions' aims and the player health study were included in this initial approach email. Session times were scheduled according to player availability, and once timings were agreed these sessions were again publicised using social media and on the University's rugby mailing lists.

Players who expressed an interest following the initial call and confirmed their availability for a session were sent further details of the session's structure. They were informed of the player health study's ethical approval and that the session would be an informal open forum discussion of their thoughts and views on the structure and content of paper and online-format questionnaires to inform the project's development.

Sessions were audio recorded and transcribed by the player-lead (EB), and then reviewed by a member of the study team also present at the sessions (MD). All data collection and processes followed good research governance principles.

The fora were not intended to be the research activity of qualitative focus groups seeking to generate new knowledge, but were designed to be player-led group discussions where players could inform researchers in designing and implementing an appropriate questionnaire and study design, which would accurately encompass player experiences and capture elements of player health that players felt needed to be better understood.

The fora

Two fora were carried out with amateur level, University rugby players. The player-lead was a current player and medical student on a research placement with limited prior exposure to the research project. Eleven players participated in the fora, with five in one session and six in the other. Ten were male and one was female.

The multidisciplinary research team produced a topic guide for the sessions with the fora lead in collaboration with an experienced qualitative researcher. Players involved in the fora did not receive any research training prior to the fora, but the player-lead had several meetings with the research team, was given an overview of the project and attended two meetings discussing the topic guide and conducting fora lead by a qualitative researcher experienced in chairing focus groups.

Players were informed that information disclosed to the group would be anonymous and treated confidentially, and that the aim of the session was questionnaire and study feedback and that there were no right or wrong responses. Further prompts and probes were used in conversation when appropriate. Key suggestions were extracted from transcripts by the player-lead and a researcher present at the sessions. Player suggestions were discussed by the research team (JN, NKA and MD) and implemented where possible by a researcher who had been present at the fora (MD). Suggestions mentioned in both fora, or repeatedly by different players within a group were prioritised for inclusion.

Results

Both fora communicated easily, freely discussing their playing experiences in order to inform the design of the study. During the review of the paper questionnaire, some players ask questions which were often answered or considered by other members of the group, before clarification was sought from the player-lead.

Players openly justified their comments and opinions with their previous playing history experiences. Group consensus was not sought or encouraged, and the groups appeared comfortable to disagree and demonstrate contrasting views. However players mostly agreed on concepts and considered issues in an iterative manner, demonstrating similar experiences and contributing to each other's ideas.

Specific consequences from the fora and ensuing key questionnaire or process alterations are outlined in Table 1.

The groups' views

Players were keen to ensure that individual questions were phrased optimally and achieved this by relating their experience and understanding of the question back to researchers and each other, such as: 'When you're asking about keyhole surgery or cartilage, have you got the option of if they've had multiple operations?' Comments such as 'I would put...' demonstrated both a consideration of the question's results, and an immersion in the project that often resulted in suggested alterations to the wording or phrasing. Players suggested where further qualification of questions was necessary and

Player's Suggestion	Change resulting from player involvement				
Use event markers	Event markers created for some questions, such as 'How ofter were you concussed?' becoming 'How often were you removed from play due to concussion?' to minimize recall bia				
Use ratios for incidence	Categories of 'every (n) games' to assess the incidence of more frequent injuries.				
Long-term outcomes associated with other sports or injuries being separate from rugby	Categories of training, playing and non-rugby created for each injury reported.				
Save and come back later	Players are now able to save information and re-enter the platform at a later point in time, to assist with sections such as Family History where this information may not be imme- diately available.				
Reorder in order of importance	Current pains and primary outcome sections were placed earlier in the questionnaire.				
Alter occupation categories	An additional category was added so players can put 'professional sports player' which previously would have had to be 'skilled manual worker'.				
Include mental health status	Questions on anxiety and depression were added as players felt that this was relevant and that it would not act as a deterrent for participation				

Table 1 F	Resulting	alterations	from	the	fora
-----------	-----------	-------------	------	-----	------

when they would be unable to answer questions: 'I think 'What is the average distance you work per week?' is going to be a tough one to work out'.

Players within both groups voiced concerns about accurately examining the pain of former players. They discussed their own current pains and the differences between acute and more long-term pain, and how pain they experienced differed between their own joints.

Based on players' experiences and interests, or areas players considered pertinent to this population and important for the game to better understand, questions were suggested as additions to the questionnaire. Players were also directly asked what they would add to the questionnaire, and suggested the availability of medical staff and support following an injury to be important to understand. The way that injuries affected your ability to work was considered both a valid marker of severity and also important to players in terms of its impact on their daily life.

There were many questions from players to researchers during the sessions. This was mostly surrounding the reasoning behind specific question choices, whether questions were standardised ways of collecting concepts or what specific information was required from a question, so players could consider if the concept was being satisfactorily captured with that question. This made researchers accountable to the group, which was an initial aim in involving players in the project.

Players were made aware in the introduction to the session that the questionnaire would be sent to former players at various amateur and professional levels. The proposed recruitment strategy for the study was discussed with players, who were able to inform researchers of potential motivators which would encourage their participation or had encouraged them to participate in previous rugby initiatives, and factors which may dissuade questionnaire completion, such as questionnaire length, boredom and extensive drop-down lists. Players discussed the differences between former and current players and often referred to the possible contrasts in answers, player experience and outlook between current and former, and amateur and professional rugby players. Within this context, some players referred to their 'dad's generation' as the older, target audience. They discussed recall bias, changes within the sport over time and a contrast-ing mentality between older and younger players.

Players regularly discussed their own experiences, and also those of other individuals in games they had played in, or elite level games they had watched. This confirms the epistemological motivation for player involvement and enables question alterations to mirror players' experiences or circumstances they had witnessed other players experiencing which would not have been accurately captured in the initial questionnaire.

Discussion

Sports participants as research partners

Current players were involved in this process as expert consultants to inform the study design, questionnaire and research agenda. This is the first occasion we are aware of sports participants being involved in informing research with their experiences. Within these sessions, players assumed the role of researchers, discussing how certain questions could be phrased to be most accurately answered, and examining the bias that may be introduced with certain terminology. Female and male clubs were approached to become involved in the fora. The uptake of female players was notably less than for males, however this may be a reflection of participation in rugby union throughout the University, with more male players involved at a College and University level providing a greater number of male attendees.

Players often asked researchers if they had considered certain elements, issues or topics. This contributed to addressing content validity in the questionnaire and also held researchers accountable to the fora, not only ensuring the effectiveness of questions and their outcomes, but also demonstrating involvement as a form of empowerment [18], for a group who may potentially be affected by the outcome of a study to have been involved in its objectives, priorities and design.

Within this consultant role, topics of questionnaire fatigue and recall bias were raised, considered and addressed by players. This resulted in researchers observing a similar process to those experienced during initial questionnaire design, and appeared to validate some of the reflection processes researchers had previously navigated. The concept of players exploring and vocalising similar considerations to researchers may be due to similar experiences between players and the research team, or similarities in the values, assumptions and priorities of players and the researchers. The impact of involvement has previously been considered to be greatest where the skills and experiences of those involved differ from the research team [8]. As researchers and players were both undergraduate and postgraduate students at the same University this similar reflection process may be a result of shared values and experience.

The pursuit of a democratic approach to healthcare provision and research, which has driven the growth of involvement in research, has been discussed as empowering healthcare users and patient representatives to contribute to the research process and hold researchers accountable in publicly funded research. Methods of actively empowering patients and members of the public have previously been suggested [31, 32], and research training for the public has been discussed as enabling members of the public to critique the research cycle more effectively and improve public confidence in articulating their sentiments during involvement [17].

During these fora, possibly due to the specific sporting population, players' prior acquaintance with each other at a University or College level, or their status as students of a high-ranking University, players easily assumed a critiquing researcher role desired within public consultations. Players were also comfortable to appraise the questionnaire and research team for prior decisions, without a specific focus of the session being their empowerment or a notion of perceived power exchange between researchers and players being stressed. Players did not receive specific research training, which has been recommended for involvement, and described as assisting with more effective critique of research by those involved [1, 17]. However, as student athletes with multiple demands on their time, researchers were concerned that an additional training session would have reduced attendance and proved too demanding for those involved. The tutorial system in which many of the student-athletes will participate during their studies may also influence their ease in discussion, as this group may be acclimatised to voicing opinions, contributing in a discussion forum and developing each other's ideas.

Training in involvement also introduces concerns of professionalization, with the potential of losing a 'lay perspective', either directly through training activities, or as a result of research knowledge and experiences gained through multiple projects [33, 34]. Researchers were concerned that players undergoing training in research methodology may lose an ideological lay perspective, which was of substantial value to researchers, more so than obtaining critique on an even footing, which may have been less applicable for this player population.

The involvement of a rugby-playing medical student as group lead who was previously acquainted with the majority of players, and had experienced limited prior exposure to the study with no official study role, but an awareness of the project and session aims, may have facilitated the groups' interaction and success. The notion that user involvement is facilitated by a perceived lack of significant operationalized power structure between the users and researchers, and that the perception of equality is vital has been previously discussed [10] and may have facilitated dialogue during these sessions.

Consultative fora in the design phase are only one method of involvement within this player health study. Current and former amateur players were involved in initial study design discussions and there are player representatives on the Study Steering Group, directly informing study processes and being involved with interpretation of study results and publication strategy decisions, in addition to future study direction.

Session Structure

A topic guide was used to broadly structure conversation and receive focused feedback on specific perceived weaknesses within the questionnaire and study design. The topic guide followed the questionnaire structure, from first impressions through to any information participants thought was omitted and should be included. This allowed players many opportunities to comment on specific questions, ensured the entire questionnaire was covered and allowed high level and specific feedback, with some inter-group consensus as players moved through the questionnaire together, in a relatively short session, which were both between one and two hours.

The session format of two fora, led by a student with a topic guide sufficiently detailed to stimulate conversation, address areas of researcher concern and provide session structure, without relying on researcher input, allowed the session to be player-led. The research team was mindful of allowing the groups to feel capable of expressing opinions and concerns without having to, or feeling that they must find faults or make alterations to the study.

A member of the study team (MD) was present as an observer for the sessions. This provided the opportunity to observe the initial reaction of players to the questionnaire. Interpretation of these responses provided insight for the quantitative study team into potential motivators and deterrents for participation upon receiving the questionnaire, which has influenced methods of recruitment by including incentives, encouraging clubs to be ambassadors for the research, involving the teams medical staff, considering timing in the season when questionnaires are sent, and considering designated times when players can participate together.

Strengths and weaknesses

The involvement of players within this questionnaire and study design was an informative process for researchers, resulting in several changes to the questionnaire (Table 1) and increased confidence of researchers that the questionnaire was approved by players and would collect player-centred information in the most precise way.

The questionnaire structure, specific questions and online functionality were altered and improved following recommendations. Questions that may be more difficult for players to answer were highlighted to researchers during the project, even if the fora or study team felt limited in scope to alter this whilst maintaining the study objectives.

As a consultative activity, the sessions were player-lead by one particular player, who had delegated responsibilities for the sessions from the study team, as opposed to being truly lead by a group of player-partners in collaborative decision-making that would be directly implemented. Researchers felt that player-collaboration was present in other aspects of the study, and that the purpose of this session was more targeted feedback, which lead to a consultative approach using a topic guide for structure.

The groups for this exercise were locally available to researchers and currently playing at an amateur level of play. A player involvement session within an older group may have been useful in order to receive any additional feedback from a group with different experiences, who had also played in the amateur era of the sport and may have a different approach to the questionnaire based on these experiences. Fora attendees being current amateur players may however benefit the study by ensuring the relevance of the research project to this population who will be influenced by the translated results, and who also form the majority of the playing population.

Further player involvement in the application of alterations and suggestions may have helped to ensure suggestions were embedded without bias from the study team, and that players maintained their authority throughout the process. However, researchers were concerned with burdening players, and the team felt that as ambassadors of involvement, player suggestions would be integrated where possible. The main reasons for noninclusion were where player suggestions may be less relevant for former players or where a suggestion would involve multiple smaller questions to be accurately captured, therefore taking too much space within the questionnaire for a secondary study objective.

The need for more critical evaluation of involvement activities has been previously discussed and efforts have been made to improve the reporting of such activities and evaluate them [35], to prevent involvement being treated as a tokenistic exercise [4]. The number of changes in the questionnaire as a result of the fora, and process of players considering research questions, limitations and outcomes using their previous experience to inform this demonstrates the worth of these sessions within this project. Players did not, however, formally evaluate the process as has been suggested previously in the literature [14, 35], although players also did not request to do so.

Key reflections on process Benefits

□ Current sports players willingly gave their time without incentives or any payment, and readily discussed the questionnaire, altering the way key variables were collected.
□ Novel thoughts were presented to researchers at a stage where they could be incorporated into the study.

□ Researchers developed increased confidence in the questionnaire's appropriateness and acceptability for players.

Difficulties

 \Box Specific funding was not available for this activity and whilst the groups were coordinated at locations convenient for players meaning researchers travelled to the groups rather than players, payment for patient involvement has been encouraged [1, 14, 17].

 \Box Training may have enabled more effective critique and improved confidence for players.

 \Box Older players may have had alternative views that were not encompassed in these fora.

□ The number of players taking part in the fora was notably lower than those approached to participate, particularly for female players. However, this process was developed to assess the feasibility of players being involved and consult players on improving the questionnaire and study processes, and not to qualitatively examine generalisable players' views.

This process in the context of previous involvement work

It has previously been suggested that involvement is limited to personal experiences by a lack of research training [16]. However, researchers within this project found increased value in this, as players bought a wealth of personal experiences and idealistic views that were not restrained by methodological or logistical considerations. These fora were coordinated with limited resource and a desire to cause minimal player inconvenience, and therefore scheduling a training element to the session may have decreased adherence and altered the demographic of those choosing to attend.

As an involvement process, these fora were not intended to answer a research question. The sessions aimed at understanding how to develop and design our study and questionnaire to be accessible and player-focused. We also sought to gain insight into some of the identified issues relating to the reporting of pain in sports players and any potential modifiable barriers to acceptance and completion within this sporting cohort. This article has summarized conversation and key resultant action points within these fora and confirmed involvement to be a research-altering process within this population.

Conclusions

This process has demonstrated the relative ease of establishing a group of amateur sports players, as a form of involvement, and the benefits of involving sports participants in athlete-focused research studies. This exercise has demonstrated that player involvement can support the foundations, goal setting and outcomes of research projects, and potentially improve the collection of sport-specific variables, as informed by player experiences.

We believe this is the first time that rugby players have been involved as expert consultants to help design, develop and review sports research in the initial stages of research development. We anticipate that this process will strengthen the recruitment of our study and ensure that the self-report questionnaire developed is relevant, playercentred and subject to minimal population-driven bias.

Acknowledgements

The authors would like to thank Sarah Drew for her assistance in developing a topic guide to facilitate focus group discussions, and player representatives at St. Anne's and St. John's Colleges, University of Oxford, for their contributions. The authors would also like to thank player representatives on our study Steering Group who continue to support and inspire this player health study.

Funding

The Arthritis Research UK Centre for Sport, Exercise and Osteoarthritis has financially supported this study and the research team (Grant reference 20194). The player health study has received partial funding from the Rugby Football Union. Neither funding body has been involved in the design, collection or interpretation of this involvement activity, or the writing of this manuscript.

Availability of data and material

Data sharing not applicable to this article as no datasets were generated or analysed during the current study.

Authors' contributions

JA, MD, JN and NA made substantial contributions to conception and design of this process. EB and MD undertook the acquisition of data and interpretation of findings. AJ has supported data interpretation, drafting and critically revising this manuscript. JC has supported the study as a player representative on the study steering group and has been involved in critically revising this manuscript as a player representative. All authors have read and approved the final manuscript.

Competing interests

The Arthritis Research UK Centre for Sport, Exercise and Osteoarthritis has financially supported this study and the research team (Grant reference 20194). The player health study has received partial funding from the Rugby Football Union. AJ has received consultancy, lecture fees and honoria from Servier, UK Renal Registry, Oxford Craniofacial Unit, IDIAP Jordi Gol, Freshfields Bruckhaus Deringer, has held advisory board positions (which involved receipt of fees) from Anthera Pharmaceuticals, INC., and received research sponsorship from ROCHE.

Consent for publication

Not applicable.

Ethics approval and consent to participate

As a public involvement process informing the questionnaire and study design, ethical approval and consent were not obtained for this process. However, the player health study this process helped inform has received ethical approval from the Central University Research Ethics Committee at the University of Oxford (MSD-IDREC-C1-2014-020).

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Author details

¹Oxford NIHR Musculoskeletal Biomedical Research Unit, University of Oxford, Oxford, UK. ²Arthritis Research UK Centre for Sport, Exercise and Osteoarthritis, University of Oxford, Oxford, UK. ³Medical Sciences Division, University of Oxford, Oxford, UK. ⁴Faculty of Health Sciences, University of Southampton, Southampton, UK. ⁵Oxford University Rugby Football Club, Oxford, UK. ⁶MRC Lifecourse Epidemiology Unit, University of Southampton, Southampton, UK.

Received: 1 September 2016 Accepted: 27 March 2017 Published online: 04 May 2017

References

- INVOLVE. Briefing Notes for Researchers: Involving the Public in the NHS, Public Health and Social Care Research. Eastleigh: INVOLVE; 2012.
- 2. Koniotou M, Halter M, Evans BA, Porter A, Gaze S. Developing a model to involve service users in a multi-centred trial of a complex intervention in pre-hospital emergency care. Emerg Med J. 2015;32:e9.1–e9.
- Adams J, Lempp H. Patients' and carers' direct involvement in musculoskeletal research and service delivery: what are the benefits and challenges? Rheumatology. 2014;53(suppl_1):keu052.
- Wright D, Foster C, Amir Z, Elliott J, Wilson R. Critical appraisal guidelines for assessing the quality and impact of user involvement in research. Health Expect. 2010;13:359–68.
- 5. NHS National Institute for Health Research. Patient and public involvement in health and social care research: A handbook for researchers. 2014.
- 6. Minogue V. Research and Development Strategy 2013–2018: Research Is Everybody's Business. 2013.
- Department of Health. Research governance framework for health and social care (Second edition, 2005). Health Soc Care Community. 2005;10:1–54.
- Staley K. Is it worth doing?' Measuring the impact of patient and public involvement in research. Res Involv Engagem. 2015;1:1–10.
- Wright DN, Corner JL, Hopkinson JB, Foster CL. The case for user involvement in research: the research priorities of cancer patients. Breast Cancer Res. 2007;9 Suppl 2:S3.
- Staniszewska S, Jones N, Newburn M, Marshall S. User involvement in the development of a research bid: barriers, enablers and impacts. Health Expect. 2007;10:173–83.
- 11. Taylor RM, Mohain J, Gibson F, Solanki A, Whelan J, Fern LA. Novel participatory methods of involving patients in research: naming and branding a longitudinal cohort study, BRIGHTLIGHT. BMC Med Res Methodol. 2015;15:20.
- Lowes L, Robling MR, Bennert K, Crawley C, Hambly H, Hawthorne K, Gregory JW. Involving lay and professional stakeholders in the development of a research intervention for the DEPICTED study. Health Expect. 2011;14:250– 60. Lowes, L., Robling, M. R., Bennert, K., Craw.

- Owens C, Farrand P, Darvill R, Emmens T, Hewis E, Aitken P. Involving service users in intervention design: a participatory approach to developing a text-messaging intervention to reduce repetition of self-harm. Health Expect. 2011;14:285–95.
- Gooberman-Hill R, Jinks C, Boucas SB, Hislop K, Dziedzic KS, Rhodes C, Burston A, Adams J. Designing a placebo device: involving service users in clinical trial design. Heal Expect. 2013;16:E100–10.
- Boote J, Wong R, Booth A. 'Talking the talk or walking the walk?' A bibliometric review of the literature on public involvement in health research published between 1995 and 2009. Heal Expect. 2012;18:44–57.
- Brett J, Staniszewska S, Mockford C, Herron-Marx S, Hughes J, Tysall C, Suleman R. Mapping the impact of patient and public involvement on health and social care research: A systematic review. Heal Expect. 2012;2012:1–14.
- Mockford C, Murray M, Seers K, Oyebode J, Grant R, Boex S, Staniszewska S, Diment Y, Leach J, Sharma U, Clarke R, Suleman R. A SHARED study-the benefits and costs of setting up a health research study involving lay coresearchers and how we overcame the challenges. Res Involv Engagem. 2016;2:8.
- Boote J, Baird W, Beecroft C. Public involvement at the design stage of primary health research: A narrative review of case examples. Health Policy (New York). 2010;95:10–23.
- Boote J, Telford R, Cooper C. Consumer involvement in health research: a review and research agenda. Health Policy (New York). 2002;61:213–36.
- McKenna J, Thomas H. Enduring injustice: a case study of retirement from professional rugby union. Sport Educ Soc. 2007;12:19–35.
- 21. Godlee F. NHS is not (yet) in crisis, but what about school rugby? BMJ. 2015;350(jan08 4):h78.
- 22. Carter M. The unknown risks of youth rugby. BMJ. 2015;2015:350.
- Kmietowicz Z. World Rugby must review rules on tackling, says public health expert. BMJ. 2015; 351(sep22_6):h5049.
- Howe PD. An Ethnography of Pain and Injury in Professional Rugby Union: The Case of Pontypridd RFC. Int Rev Sociol Sport. 2001;36:289–303.
- Liston K, Reacher D, Smith A, Waddington I. Managing Pain and Injury in Non-Elite Rugby Union and Rugby League: a case study of players at a British University. Sport Soc. 2006;9(3):388–402.
- Flicker S, O'Campo P, Monchalin R, Thistle J, Worthington C, Masching R, Guta A, Pooyak S, Whitebird W, Thomas C. Research done in 'A good way: The importance of indigenous elder involvement in HIV community-based research. Am J Public Health. 2015;105:1149–54.
- 27. Chang AB, Taylor B, Masters IB, Laifoo Y, Brown ADH. Indigenous healthcare worker involvement for Indigenous adults and children with asthma. Cochrane Database Syst Rev. 2007;1:5.
- Beardmore AL, Handcock PJ, Rehrer NJ. Return-to-play after injury: practices in New Zealand rugby union. Phys Ther Sport. 2005;6:24–30.
- 29. Hollis SJ, Stevenson MR, McIntosh AS, Shores EA, Finch CF. Compliance with return-to-play regulations following concussion in Australian schoolboy and community rugby union players. Br J Sport Med. 2012;46:735–40.
- 30. Murphy P, Waddington I. Are Elite Athletes Exploited? Sport Soc. 2007;10:239–55.
- 31. Piper S. How To Empower Patients And Involve The Public. Nurs Stand. 2014;29:37-41.
- 32. Leppo A, Perälä R. User involvement in Finland: the hybrid of control and emancipation. J Health Organ Manag. 2009;23:359–71.
- 33. Cassidy J. Why patient representation might harm science? Breast Cancer Res. 2007;9 Suppl 2:S4.
- 34. Dudley L, Gamble C, Allam A, Bell P, Buck D, Goodare H, Hanley B, Preston J, Walker A, Williamson P, Young B, Rose D, Selby J, Lipstein S, Noe R, Tews M, Dachner AM, Staley K, Gamble C, Dudley L, Allam A, Bell P, Goodare H, Hanley B, Buck D, Gamble C, Dudley L, Preston J, Hanley B, Williamson P, et al. A little more conversation please? Qualitative study of researchers' and patients' interview accounts of training for patient and public involvement in clinical trials. Trials. 2015;16:190.
- 35. Staniszewska S, Brett J, Mockford C, Barber R. The GRIPP checklist: strengthening the quality of patient and public involvement reporting in research. Int J Technol Assess Health Care. 2011;27:391–9.

Submit your next manuscript to BioMed Central and we will help you at every step:

- We accept pre-submission inquiries
- Our selector tool helps you to find the most relevant journal
- We provide round the clock customer support
- Convenient online submission
- Thorough peer review
- Inclusion in PubMed and all major indexing services
- Maximum visibility for your research

Submit your manuscript at www.biomedcentral.com/submit

